Author’s response to reviews

Title: Low-risk susceptibility alleles in 40 human breast cancer cell lines

Authors:

Muhammad Riaz (m.riaz@erasusmc.nl)
Fons Elstrodt (f.elstrodt@bioke.com)
Antoinette Hollestelle (a.hollestelle@erasusmc.nl)
Abbas Dehghan (a.dehghan@erasusmc.nl)
Jan G M Klijn (j.g.m.klijn@erasusmc.nl)
Mieke Schutte (a.schutte@erasusmc.nl)

Version: 6 Date: 16 July 2009

Author’s response to reviews: see over
Dear Dr. Kaiser

Thank you for accepting our manuscript for publication in BMC Cancer. We have formatted and uploaded the final manuscript today. As per requirements of the journal, we have performed the major and minor revisions to the format of our manuscript. In the final manuscript we have split up table 1 into two tables; table 1 & 2 and we have cited these tables in the text accordingly. We believe that the manuscript now is ready for publication in BMC Cancer.

Yours sincerely,

Mieke Schutte